Author's response to reviews

Title: Performance Feedback: An exploratory study to examine the acceptability and impact for interdisciplinary primary care teams

Authors:

Sharon Johnston (sjohnston@bruyere.org)
Green Michael (michael.green@dfm.queensu.ca)
Patricia Thille (pthille@uottawa.ca)
Colleen Savage (savagec@queensu.ca)
Lynn Roberts (roberts@queensu.ca)
Grant Russell (Grant.Russell@med.monash.edu.au)
William Hogg (whogg@uottawa.ca)

Version: 5 Date: 21 March 2011

Author's response to reviews: see over
Dear Editor,

Re: Manuscript Re-Submission

Thank you for the opportunity to re-submit our manuscript titled, Performance Feedback: An exploratory study to examine the acceptability and impact for interdisciplinary primary care teams, for consideration by BMC Family Practice. We have reviewed the manuscript available online and have made the last minor changes necessary including:

- Updating the contact details for Dr. Grant Russell to reflect his current coordinates.
- In the version we uploaded previously, arrows exist in Table 1 to indicate a range, however, in the proof these arrows have been replaced by other, incomprehensible symbols. Arrows have been removed and replaced with the word “to” to avoid further formatting errors.

The following authors have contributed to the writing of this manuscript and approved the final version: Drs. Johnston, Green, Thille, Savage, Roberts, Russell and Hogg.

I confirm that all authors have seen and agree with the contents of this manuscript and confirm that the work has not been submitted or published elsewhere in whole or in part.

This manuscript is submitted in accordance with the regulations as defined by the BMC Family Practice editorial policies.

Please do not hesitate to contact me if any further information is required.

Sincerely,

Dr. Sharon Johnston – Corresponding author
Assistant Professor
Department of Family Medicine
University of Ottawa
CT Lamont Primary Health Care Research Centre
Élisabeth Bruyère Research Institute
43 Bruyère Street
Ottawa, On, K1N 5C8
Tel: (613) 562-0050 ext.1599